Author’s response to reviews

Title: Potentiating and synergistic effect of grape fruit juice on the antioxidant and antiinflammatory activity of aripiprazole against hydrogen peroxide induced oxidative stress in mice

Authors:
Seema Zargar (szargar@ksu.edu.sa)
Tanveer Wani (twani@ksu.edu.sa)
Abdul-Rahman Al-Majed (tanykash@yahoo.co.in)

Version: 2 Date: 05 Mar 2017

Author’s response to reviews:

The grape fruit juice used in the study has been characterized as per the suggestion.

1. The introduction part (paragraph 2) has been accordingly revised.

2. Table 1 has been added to the manuscript and the earlier table 1 has been renamed as table 2.

3. Flavonoid analysis: section has been included in the experimental part

4. Results from the GFJ analysis have been included in the results section.

5. Accordingly paragraph 1 and 3 of discussion have been modified